Author's response to reviews

Title: Which family physician should I choose? The analytic hierarchy process approach in the selection of a family physician

Authors:

Emel Kuruoglu (emel.kuruoglu@deu.edu.tr)
Dilek Guldal (dilek.guldal@deu.edu.tr)
Vildan Mevsim (vildan.mevsim@deu.edu.tr)
Tolga Gunvar (tolga.gunvar@deu.edu.tr)

Version: 2
Date: 16 April 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

Thank you for your revision comments. We have made changes and upload the revised manuscript. The revisions we have made are:

- We have included the title page at the front of the manuscript and changed the information of the title page as you comment.
- We provide abstract in the manuscript.
- We have put line and page number.
- We put competing interests section.
- We required authors contribution section.
- We put the figures in the main manuscript.

With my best regards,

Vildan Mevsim, Assoc.Prof, MD, PhD
Institution: Dokuz Eylul University Medical Faculty Department of Family Medicine
Address: Dokuz Eylül Üniversitesi Tıp Fakültesi Aile Hekimliği AD. 35340 İnciraltı İzmir TURKEY
Tel: +90 532 2360044
Fax: +90 4124959
E-mail: vildan.mevsim@deu.edu.tr